Bisphosphonates to reduce bone fractures in stage 3B+ chronic kidney disease: a propensity score-matched cohort study

Danielle E Robinson, 1[†] M Sanni Ali, 1,2[†] Victoria Y Strauss, 1* Leena Elhussein, 1
Bo Abrahamsen, 3,4 Nigel K Arden, 5,6
Yoav Ben-Shlomo, 7 Fergus Caskey, 8,9 Cyrus Cooper, 1,6
Daniel Dedman, 10 Antonella Delmestri, 1
Andrew Judge, 1,11,12 Muhammad Kassim Javaid 1,6
and Daniel Prieto-Alhambra 1,13

- ¹Nuffield Department of Orthopaedics, Rheumatology and Musculoskeletal Sciences, National Institute for Health Research (NIHR) Biomedical Research Centre, University of Oxford, Oxford, UK
- ²Faculty of Epidemiology and Population Health, Department of Non-communicable Disease Epidemiology, London School of Hygiene & Tropical Medicine, London, UK
- ³Open Patient data Explorative Network (OPEN), Department of Clinical Research, University of Southern Denmark, Odense, Denmark
- ⁴Department of Medicine, Holbæk Hospital, Holbæk, Denmark
- ⁵Arthritis Research UK Sports, Exercise and Osteoarthritis Centre, University of Oxford, Oxford, UK
- ⁶Medical Research Council Lifecourse Epidemiology Unit, University of Southampton, Southampton General Hospital, Southampton, UK
- ⁷Population Health Sciences, University of Bristol, Bristol, UK
- ⁸School of Social and Community Medicine, University of Bristol, Bristol, UK ⁹UK Renal Registry, Bristol, UK
- ¹⁰Clinical Practice Research Datalink, Medicines and Healthcare products Regulatory Agency, London, UK
- ¹¹Musculoskeletal Research Unit, Translational Health Sciences, Bristol Medical School, University of Bristol, Bristol, UK
- ¹²National Institute for Health Research (NIHR) Bristol Biomedical Research Centre (BRC), University Hospitals Bristol NHS Foundation Trust, University of Bristol, Southmead Hospital, Bristol, UK
- ¹³Grup de Recerca en Malalties Prevalents de l'Aparell Locomotor (GREMPAL) Research Group and Centro de Investigación Biomédica en Red Fragilidad y Envejecimiento Saludable (CIBERFes), University Institute for Primary Care Research (IDIAP) Jordi Gol, Universitat Autonoma de Barcelona and Instituto de Salud Carlos III, Barcelona, Spain

^{*}Corresponding author victoria.strauss@csm.ox.ac.uk †Joint first authors

Declared competing interests of authors: Bo Abrahamsen reports research grants from Union Chimique Belge (UCB S.A. Brussels, Belgium) and Novartis International AG (Basel, Switzerland). Nigel K Arden reports personal fees from Flexion Therapeutics, Inc. (Burlington, MA, USA), Freshfields Bruckhaus Deringer LLP (London, UK), Janssen Pharmaceutica (Beerse, Belgium), Merck Group (Darmstadt, Germany) and Regeneron Pharmaceuticals Inc. (Eastview, NY, USA). Cyrus Cooper reports personal fees from the Alliance for Better Bone Health Amgen (Amgen Inc., Thousand Oaks, CA, USA), Eli Lilly and Company (Indianapolis, IN, USA), GlaxoSmithKline plc (London, UK), Medtronic plc (Dublin, Ireland), Merck, Novartis, Pfizer Inc. (New York City, NY, USA), F. Hoffmann-La Roche AG (Basel, Switzerland), Servier Laboratories (Neuilly-sur-Seine, France), Takeda Pharmaceutical Company Ltd (Tokyo, Japan) and UCB. Andrew Judge is a subpanel member of the National Institute for Health Research (NIHR) Programme Grants for Applied Research panel, has received consultancy fees from Freshfields Bruckhaus Deringer and has held advisory board positions (which involved receipt of fees) at Anthera Pharmaceuticals, Inc. (Hayward, CA, USA). M Kassim Javaid reports research support and speaker fees from Amgen. Daniel Prieto-Alhambra reports research grants from UCB, Amgen and Servier Laboratories, and departmental (not personal) speaker and consultancy fees from UCB and Amgen. He has also been a member of the NIHR Health Technology Assessment Clinical Evaluation and Trials panel since November 2017.

Published March 2021 DOI: 10.3310/hta25170

Plain English summary

Bisphosphonates for bone fractures in CKD

Health Technology Assessment 2021; Vol. 25: No. 17

DOI: 10.3310/hta25170

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

Rationale and aims

Bisphosphonates are used to prevent fractures in people with fragile bones. People with chronic kidney disease have a high risk of fracturing, but the safety and effectiveness of bisphosphonates in severe chronic kidney disease is unclear. The aim of this study was to assess the benefits (e.g. bone strength improvement and fracture prevention) and the risks of unwanted effects associated with bisphosphonates for people with moderate to severe chronic kidney disease.

Methods

Anonymised primary and secondary care electronic medical records data from the UK NHS were used, as well as a Danish equivalent that included bone density scans.

Anyone in these databases with a measure of reduced kidney function that suggested moderate to severe chronic kidney disease was eligible, which was > 220,000 people from the UK. Over 20,000 of them used bisphosphonates. Bisphosphonate users were matched to non-users with similar age, sex and other characteristics.

Results

Bisphosphonate users had a 12% higher risk of their chronic kidney disease getting worse than non-users. Their risks of other side effects, such as acute kidney injuries and gastrointestinal problems, did not change.

Bisphosphonate users had a 25% higher risk of fractures than non-users in the UK database, probably because the matching methods did not create similar-enough groups of users and non-users. However, it was found that bisphosphonate improved bone density in the Danish database. Bone density is a proxy for bone strength, so better bone density should mean fewer fractures.

Conclusions

These results suggest that bisphosphonate therapy may make moderate to severe chronic kidney disease worse. More studies are needed on how bisphosphonates affect milder chronic kidney disease.

Bisphosphonates were associated with better bone strength, but it could not be demonstrated that they reduced fracture risk. More data are required, probably from a placebo-controlled trial, to determine whether or not bisphosphonates prevent fractures in people with moderate to severe chronic kidney disease and whether or not this is worth the risk of their chronic kidney disease worsening.

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 3.370

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, the Cochrane Library and Clarivate Analytics Science Citation Index.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

This report

The research reported in this issue of the journal was funded by the HTA programme as project number 14/36/02. The contractual start date was in December 2015. The draft report began editorial review in October 2018 and was accepted for publication in March 2019. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2021. This work was produced by Robinson et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Scientific Adviser (Evidence Use), Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk